Author's response to reviews

Title: Characterization of ERK Docking Domain Inhibitors that Induce Apoptosis by Targeting Rsk-1 and Caspase-9

Authors:

Sarice R Boston (ssmit022@umaryland.edu)
Rahul Deshmukh (rdesh003@umaryland.edu)
Scott Strome (sstrome@smail.umaryland.edu)
U. Deva Priyakumar (deva@iiit.ac.in)
Alexander D. MacKerell Jr. (amackere@rx.umaryland.edu)
Paul Shapiro (pshapiro@rx.umaryland.edu)

Version: 4 Date: 6 January 2011

Author's response to reviews: see over

Title: Characterization of ERK Docking Domain Inhibitors that Induce Apoptosis by Targeting Rsk-1 and Caspase-9

Journal: BMC Cancer

MS ID: 1774508556440059

**Formatting changes to be made:**

---

**Major revisions**

1. We have added the affiliations including Department, Institute, (Street), City, (Postal code), Country as requested.

**Minor revisions**

1. We have corrected the Email addresses for all the authors on the title page according to the following format: AB: abcd@institution.ac.uk

2. The word ‘Introduction’ has been renamed 'Background'.

3. All highlighting/tracking have been removed from the manuscript.

4. Tables- The visible vertical lines have been removed from the tables

5. The figures have been cropped as closely as possible to minimise white space around the image.

6. Figure 7 has been removed from the submission system.